Author's response to reviews

Title: The role of the fat mass and obesity associated gene (FTO) in breast cancer risk

Authors:

virginia kaklamani (v-kaklamani@northwestern.edu)
Nengjun Yi (nyi@ms.soph.uab.edu)
Maureen Sadim (maui@northwestern.edu)
Kalliopi Siziopikou (p-siziopikou@northwestern.edu)
Kui Zhang (KZhang@ms.soph.uab.edu)
Yianfei Xu (yanfei-xu@northwestern.edu)
Sarah Tofilon (stofilon@gmail.com)
Surbhi Agarwal (surbhi1992@yahoo.com)
Boris Pasche (drpasche@gmail.com)
Christos Mantzoros (cmantzor@bidmc.harvard.edu)

Version: 2 Date: 21 March 2011

Author's response to reviews:

Major and minor revisions have been done according to the format of the journal